Author's response to reviews

Title: Use of peers and community lay persons improves six-week postnatal clinic (PNC) follow-up and Early Infant HIV Diagnosis (EID) in urban and rural health units in Uganda: A two-year implementation study

Authors:

Zikulah Namukwaya (znamukwaya@mujhu.org)
Linda Barlow-Mosha (lbarlow@mujhu.org)
Peter Mudiope (mudiopek@yahoo.com)
Adeodata Kekitiinwa (akekitiinwa@baylor-uganda.org)
Joyce N Matovu (jmatovu@mujhu.org)
Ezra Musingye (emusingye@yahoo.com)
Jane N Sebaggala (jssebaggala@mujhu.org)
Teopista Nakyanzi (tnakyanzi@mujhu.org)
Jubilee Abwooli (jubileejohn@yahoo.co.uk)
Dorothy O Mirembe (dmirembe@mujhu.org)
Juliane Etima (jetima@mujhu.org)
Edward Bitarakwate (ebitarakwate@pedaids.org)
Mary G Fowler (mgfowler@mujhu.org)
Philippa M Musoke (pmusoke@mujhu.org)

Version: 2 Date: 8 February 2015

Author's response to reviews: see over
To the Editor
BMC Public Health

08 February 2015

Dear Ms. Jane M. Dalumpines
Journal Editorial Office
BioMed Central

Please see the changes as requested by you for the manuscript entitled, “Use of peers and community lay persons improves six-week postnatal clinic (PNC) follow-up and Early Infant HIV Diagnosis (EID) in urban and rural health units in Uganda: A two-year implementation study,”

1. I have added a cover page and all authors affiliation and addresses includes
2. I have included on the same page the full address of the submitting author
3. I have added a list of abbreviations used on page 2 of the cover page (not clear where this should go at this time). Please also note that the expanded form of all the abbreviations is listed in the text where the abbreviation id first used.
4. I have deleted the author listing from the manuscript, and left this on the cover page
5. I have changed Introduction to background in the main manuscript
6. I have added line numbering to the whole document, minus the cover page
7. I have removed page breaks

I have attempted to respond to your entire request at this time,

Thank you,

Zikulah Namukwaya, MBChB, MPH
MU-JHU CARE LTD
Old Mulago Hill road
P.O Box 23591,
Kampala, Uganda
znamukwaya@mujhu.org